6611 Network Processor

GX27-3909-02

Operations Pocket Guide

Operations Pocket Guide



Third Edition (May 1993)

Note

Before using this information and the product it supports, be sure to read the general information under "Notices" on page xi.

This edition applies to the IBM 6611 Network Processor Models 120, 140, and 170.

Publications are not stocked at the address given below. Order publications through your IBM representative or the IBM branch office serving your locality.

A form for readers' comments is provided at the back of this publication. If the form has been removed, address your comments to:

IBM Corporation Department E15 P.O. Box 12195 Research Triangle Park, NC 27709-9990 U.S.A.

When you send information to IBM, you grant IBM a nonexclusive right to use or distribute the information in any way it believes appropriate without incurring any obligation to you.

© Copyright International Business Machines Corporation 1992, 1993. All rights reserved. Note to U.S. Government Users — Documentation related to restricted rights — Use, duplication or disclosure is subject to restrictions set forth in GSA ADP Schedule Contract with IBM Corp.

Contents

Notices	xi
Electronic Emission Notices	cii
Canadian Department of Communications	
compliance statement	111
Avis de conformité aux normes du ministère	
des Communications du Canada	i i i
Notice to Users in the United Kinadom	111
Radio Protection for Germany	iv
International Electrotechnical Commission	
(IEC) Statement	iv
Japanese Voluntary Control Council for	
Interference (VCCI) Statement	iv
Korean Communications Statement	٠v
Safety	vi
Translated Safety Information	vi
Traduction des consignes de sécurité xv	/11
Traduction des Consignes de sécurité xv	/ii
Vertaalde veiligheidsvoorschriften	/11
Traduzione delle informazioni di sicurezza xvi	111
Übersetzte Sicherheitshinweise	iii
Tradução das Informações Sobre Segurança x	ix
Informações de Segurança Traduzidas	ix
Información de Seguridad Traducida	κx
Oversatt sikkerhetsinformasjon	x
Suomenkieliset turvaohjeet	x
Oversatte sikkerhedsforskrifter	xi
Säkerhetsinformationx	xi
Trademarks xx	iii
About This Book	٧
Who Should Use This Book	vi
Part I: IBM 6611 Network Processor	
Model 120	
Chapter 1 Description of the IBM 6611 Model	

120 1-1 Parts of the Model 120 1-3	onupier	••	000				.					•	•••				•		
Parts of the Model 120 1-3	120 .																		1-1
	Parts of t	he	Мос	let	120)		•	•	• •	 •	•		•	•	•	•	•	1-3

© Copyright IBM Corp. 1992, 1993

External Device Connectors	•	•	•	•	•	•	•	•	•		1-5
Chapter 2. Using the Model 120											2-1
Setting the Key Mode Switch											2-1
Starting Model 120 Operation											2-3
Stopping Model 120 Operation											2-4
Reading the 3-Digit Display											2-4
Using the Reset Button											2-5
Loading a 3.5-Inch Diskette											2-6
Unloading a 3.5-Inch Diskette .	•	•	•	•		•	•	•	•	•	2-7
Chapter 3. Relocating the Mode	1	2	0								3-1

Part II: IBM 6611 Network Processor Model 140

Chapter 4. Description of the IBM 6611 Model	
140	4-1
Parts of the Model 140	4-3
External Device Connectors	4-5
Chapter 5. Using the Model 140	5-1
Key Mode Switch Positions	5-1
Starting Model 140 Operation	5-3
Stopping Model 140 Operation	5-4
Reading the 3-Digit Display	5-4
Using the Reset Button	5-5
Loading a 3.5-Inch Diskette	5-6
Unloading a 3.5-Inch Diskette	5-6
Changing the Pedestal on a Model 140	5-7
From Horizontal to Vertical Position	5-7
From Vertical to Horizontal Position	5-9
Changing the Position of the 3-Digit Display . 5	-13
Chapter 6. Relocating the Model 140	6-1

Part III: IBM 6611 Network Processor Model 170

Chapter 7. Description of the IBM 6611 Model	7-1
Parts of the Model 170	7-3
External Device Connectors	7-6
Chapter 8. Using the Model 170	8-1
Setting the Key Mode Switch	8-1
Starting Model 170 Operation	8-4
Stopping Model 170 Operation	8-5
Reading the 3-Digit Display	8-5
Using the Reset Button	8-6
Loading a 3.5-Inch Diskette	8-7
Unloading a 3.5-Inch Diskette	8-8
Chapter 9. Relocating the Model 170	9-1
Part IV: Common Information for Models 120, 140, and 170	
Chapter 10. ASCII Terminal Installation and Service Tasks	0-1

Service Tasks	10-1
Chapter 11. Using a Modem for Remote IBM	
Service	11-1
Chapter 12. Problem Determination	12-1
The Functions of the Diagnostic Program	12-1
Diagnostic Routines	12-1
Running Concurrent Mode Diagnostics	12-1
Running Concurrent Mode Diagnostics on a	
Direct-Attached ASCII terminal	12-2

Running Concurrent Mode Diagnostics on a	
Modem-Attached ASCII terminal	12-3
Running Concurrent Mode Diagnostics via	
the IP Network	12-3
Service Aids	12-4

Service Hints	12-4
Display Previous Diagnostic Results	12-4
Display or Change Configuration or Vital	
Product Data (VPD)	12-5
Display or Change Diagnostic Test List	12-5
Diskette Media	12-5
Reset Bootlist	12-6
Chapter 13. Cable Label	13-1
Understanding the Cable Label	13-1
Explanation of Label Fields	13-3
Appendix A. 3-Digit Display Codes	A-1
Reading the 3-Digit Display	A-1
Reading Flashing 888 Numbers	A-3
Reading Type 102 Messages	A-3
Reading Type 103 Messages	A-4
Reading Other Message Types	A-6
Crash Codes for Type 102 Messages	A-7
Dump Status Codes for Type 102 Messages	A-8
Error and Status Codes	A-9
Appendix B. Ordering Keys	B-1
List of Abbreviations	. X-1

Figures

1-1.	Model 120	. 1-1
1-2.	Parts of the Model 120	. 1-3
1-3.	Rear View of the Model 120	. 1-5
2-1.	Key Mode Switch Positions	. 2-1
2-2.	Front View of the Model 120	. 2-3
2-3.	Location of the Reset Button	. 2-5
2-4.	Loading a Diskette	. 2-6
2-5.	Unloading a Diskette	. 2-7
4-1.	Example of Vertical and Horizontal	
	Positions for the Model 140	4-1
4-2.	Model 140 with Rear Cover Removed	4-3
4-3.	Front View of the Model 140 with the	
	3-Digit Display Hinged Lens Open	. 4-5
4-4.	Rear of the Model 140 Serial Ports	
	and Adapter Slots	. 4-5
5-1.	Key Mode Switch Positions	. 5-1
5-2 .	Front View of the Model 140	. 5-3
5-3.	Location of the Reset Button	. 5-5
5-4.	Loading a Diskette	. 5-6
5-5.	Unloading a Diskette	. 5-6
5-6.	Attaching the Pedestal for the Vertica	1
	Position	. 5-8
5-7.	Removing the Pedestal in the	
	Vertical Position	5-10
5-8.	Attaching the Pedestal for the	
	Horizontal Position	5-11
5-9.	3-Digit Display in the Vertical	
	Position with the Hinged Lens	
	Closed	5-13
5-10.	3-Digit Display in the Vertical	
	Position with the Hinged Lens Open	5-13
5-11.	3-Digit Display in the Horizontal	
	Position with the Hinged Lens Open	5-14
5-12.	3-Digit Display in the Horizontal	
	Position with the Hinged Lens	
	Closed	5-14
5-13.	3-Digit Display in the Vertical	
	Position with the Hinged Lens Open	5-15

5-14.	3-Digit Display in the Horizontal	
•	Position with the Hinged Lens Open	5-16
5-15.	3-Digit Display in the Horizontal	
	Position with the Hinged Lens	
	Closed	5-17
7-1.	Model 170	. 7-2
7-2.	Front View of the Model 170	. 7-3
7-3.	Side View of the Model 170	. 7-3
7-4.	Rear View of the Model 170	. 7-4
8-1.	Key Mode Switch Positions	. 8-1
8-2.	Front View of the Model 170	. 8-4
8-3.	Location of the Reset Button	. 8-6
8-4.	Loading a Diskette	. 8-7
8-5.	Unloading a Diskette	. 8-8
10-1.	Connecting a Terminal to an EIA 232	
	Serial Port for Model 120	10-6
10-2.	Connecting a Terminal to an EIA 232	
	Serial Port for Models 140 and 170	10-7
10-3.	Terminal Interposer Connectors	10-8
10-4.	10-to-25-Pin Converter Cable	
	Connectors	10-9
A-1.	An Error Code Displayed on the	
	Model 120	A-1
A-2.	An Error Code Displayed on the	
	Model 140	A-1
A-3.	An Error Code Displayed on the	
	Model 170	A-2
		· · · -

Tables

2-1.	Summary of the Possible Operations
	for Each Key Mode Switch Position . 2-1
5-1.	Summary of the Possible Operations
	for Each Key Mode Switch Position . 5-1
8-1.	Summary of Possible Operations for
	Each Key Mode Switch Position 8-2
10-1.	Communication Setup Attributes for
	the ASCII Terminal or Emulator 10-2

10-2.	Communication Setup Attributes for	
	the ASCII Terminal or Emulator	10-4
10-3.	ASCII Terminal Connection to an EIA	
	232 Serial Port for Model 120	10-6
10-4.	ASCII Terminal Connection to an EIA	
	232 Serial Port for Models 140 and	
	170	10-7
10-5.	Terminal Interposer Pin Wiring	10-8
10-6.	10-to-25 Pin Converter Cable Pin	
	Assignments	10-9

x 6611 Operations Pocket Guide

Notices

References in this publication to IBM products, programs, or services do not imply that IBM intends to make these available in all countries in which IBM operates. Any reference to an IBM product, program, or service is not intended to state or imply that only IBM's product, program, or service may be used. Any functionally equivalent product, program, or service that does not infringe any of IBM's intellectual property rights may be used instead of the IBM product, program, or service. Evaluation and verification of operation in conjunction with other products, except those expressly designated by IBM, is the user's responsibility.

IBM may have patents or pending patent applications covering subject matter in this document. The furnishing of this document does not give you any license to these patents. You can send license inquiries, in writing, to the IBM Director of Commercial Relations, IBM Corporation, Purchase, NY 10577.

Electronic Emission Notices

Federal Communications Commission (FCC) Statement

Note: This equipment has been tested and found to comply with the limits for a Class A digital device, pursuant to Part 15 of the FCC Rules. These limits are designed to provide reasonable protection against harmful interference when the equipment is operated in a commercial environment. This equipment generates, uses, and can radiate radio frequency energy and, if not installed and used in accordance with the instruction manual, may cause harmful interference to radio communications. Operation of this equipment in a residential area is likely to cause harmful interference, in which case the user will be required to correct the interference at his own expense.

Properly shielded and grounded cables and connectors must be used in order to meet FCC emission limits. IBM is not responsible for any radio or television interference caused by using other than recommended cables and connectors or by unauthorized changes or modifications to this equipment. Unauthorized changes or modifications could void the user's authority to operate the equipment.

This device complies with Part 15 of the FCC Rules. Operation is subject to the following two conditions: (1) this device may not cause harmful interference, and (2) this device must accept any interference received, including interference that may cause undesired operation.

Canadian Department of Communications compliance statement

This equipment does not exceed Class A limits per radio noise emissions for digital apparatus, set out in the Radio Interference Regulation of the Canadian Department of Communications. Operation in a residential area may cause unacceptable interference to radio and TV reception requiring the owner or operator to take whatever steps are necessary to correct the interference.

Avis de conformité aux normes du ministère des Communications du Canada

Cet équipement ne dépasse pas les limites de Classe A d'émission de bruits radioélectriques pour les appareils numériques, telles que prescrites par le Règlement sur le brouillage radioélectrique établi par le ministère des Communications du Canada. L'exploitation faite en milieu résidentiel peut entraîner le brouillage des réceptions radio et télé, ce qui obligerait le propriétaire ou l'opérateur à prendre les dispositions nécessaires pour en éliminer les causes.

Notice to Users in the United Kingdom

The IBM 6611 Network Processor is manufactured to the International Safety Standard IEC950 and as such is approved in the United Kingdom under the General Approval number NS/G/1234/J/100003.

The network adapter interfaces, housed within the 6611, are approved separately. Each one has its own independent approval number.

These interface adapters, supplied by IBM, do not contain excessive voltages. An excessive voltage is one which exceeds 70.7v peak a.c or 120v d.c. They interface with the IBM 6611 using Safe Extra Low Voltages only. In order to maintain the independent approval of the IBM adapters, it is essential that other optional cards, not supplied by IBM, do not use mains voltages or any other excessive voltages. Seek advice from a competent engineer before installing other adapters not supplied by IBM.

Radio Protection for Germany

Instructions to User: Properly shielded and grounded cables and connectors must be used for connection to peripherals in order to meet German emission limits.

Proper cables are available from IBM authorized dealers.

Order Information: For new orders, contact an IBM authorized sales representative. For replacement orders, contact an IBM authorized service representative.

International Electrotechnical Commission (IEC) Statement

This product has been designed and built to comply with (IEC) Standard 950.

Japanese Voluntary Control Council for Interference (VCCI) Statement

This equipment is Class 1 Equipment (information equipment to be used in commercial and industrial districts) which is in conformance with the standard set by Voluntary Control for Interference by Data Processing Equipment and Electronic Office Machines (VCCI) with an aim to prevent radio interference in commercial and industrial districts. This equipment could cause interference to radio and television receivers when used in and around residential districts. Please handle the equipment properly according to the instruction manual. 情報処理装置等電波障害自主規制協議会 (VCCI)表示

電波障害自主規制届出装置の記述

この装置は、第一種情報装置(商工業地域において 使用されるべき情報装置)で商工業地域での電波障 害防止を目的とした情報処理装置等電波障害自主規 制協議会(VCCI)基準に適合しております。 従って、住宅地域またはその隣接した地域で使用す ると、ラジオ、テレビジョン受信機等に受信障害を 与えることがあります。

関連マニュアルに従って正しい取り扱いをしてくだ さい。

Korean Communications Statement

Please note that this device has been approved for business use with regard to electromagnetic wave interference. If you find this is not suitable for your use, you may exchange it for one designated for non-business purposes.

대한민국 통신문

이 / // /는 업무용으로 전자×장해/4정을 받은 / // 이오니 ×매자 또는 사용자는 이 점을 주의하시/ | 바라며, 만약 잘못 구입하였을 때에는 구입한 곳에서 비업무용으로 교환하시/ | 바랍니다.

Safety

Three types of safety notices are printed throughout this guide:

DANGER	Warns you of conditions or procedures that can result in death or severe personal injury.
CAUTION	Warns you of conditions or procedures that can cause personal injury that are neither lethal nor extremely hazardous.
Warning	Warns you of conditions or procedures that can cause damage to machines, equipment, or programs.

Translated Safety Information

Some countries require the safety information contained in publications to be presented in their national languages. The *IBM 6611 Network Processor Translated Safety Information*, GA27-3954, contains translations of the caution and danger notices and other safety information that appear in these English-language publications:

- IBM 6611 Network Processor Installation and Service Guide, GA27-3941
- IBM 6611 Network Processor Operations Pocket Guide, GX27-3909.

Refer to the booklet any time you do not understand the safety information in those English-language publications.

Traduction des consignes de sécurité

Certains pays exigent que les consignes de sécurité incluses dans les publications soient traduites dans leur langue. Dans le manuel *Traductions de consignes de sécurité pour le processeur de réseau IBM 6611*, référence GA27-3954, vous trouverez les consignes Danger, Attention et autres, contenues dans les publications en anglais ci-après :

- 6611 Network Processor Installation and Service Guide, GA27-3941
- 6611 Network Processor Operations Pocket Guide, GX27-3909.

Consultez ce livret quand vous ne comprenez pas des informations liées à sécurité présentes dans ces publications en anglais.

Traduction des Consignes de sécurité

Certains pays exigent que les consignes de sécurité figurant dans les manuels soient écrites dans leur langue nationale. La brochure 6611 Network Processor Translated Safety Information, numéro GA27-3954, contient les traductions des consignes Danger et Attention figurant dans les manuels anglais suivants:

- 6611 Network Processor Installation and Service Guide, GA27-3941
- 6611 Network Processor Operations Pocket Guide, GX27-3909.

Veuillez consulter cette brochure lorsque vous ne comprenez pas les consignes de sécurité dans ces manuels.

Vertaalde veiligheidsvoorschriften

In sommige landen bestaat de verplichting de veiligheidsvoorschriften in handleidingen in de eigen nationale taal op te stellen. De 6611 Network Processor Translated Safety Information, bestelnummer GA27-3954, bevat vertalingen van de Let op- en Gevaarberichten die in de volgende Engelstalige publicaties voorkomen:

- 6611 Network Processor Installation and Service Guide, GA27-3941
- 6611 Network Processor Operations Pocket Guide, GX27-3909.

U moet deze brochure gebruiken wanneer u de veiligheidsvoorschriften in deze Engelstalige publicaties niet begrijpt.

Traduzione delle informazioni di sicurezza

In alcuni paesi le informazioni di sicurezza, contenute nelle pubblicazioni in lingua inglese, devono essere tradotte in lingua nazionale. La pubblicazione 6611 Network Processor Translated Safety Information, GA27-3954, offre la traduzione di avvisi di pericolo e di attenzione ed altre informazioni relative alla sicurezza, contemplate nelle seguenti pubblicazioni in lingua inglese:

- 6611 Network Processor Installation and Service Guide, GA27-3941
- 6611 Network Processor Operations Pocket Guide, GX27-3909.

Consultare questo opuscolo qualora le informazioni di sicurezza, contenute nelle suddette pubblicazioni, non vengano comprese.

Übersetzte Sicherheitshinweise

In Deutschland müssen Sicherheitshinweise, die in einer Veröffentlichung enthalten sind, in deutscher Sprache vorliegen. Das Handbuch 6611 Network Processor Translated Safety Information, IBM Fomr GA27-3954, enthält die Übersetzungen der Vorsichts- und Achtungshinweise sowie anderer Sicherheitshinweise, die in den folgenden Veröffentlichungen vorkommen:

 6611 Network Processor Installation and Service Guide, IBM Form GA27-3941 • 6611 Network Processor Operations Pocket Guide, IBM Form GX27-3909.

Dieses Buch sollte verwendet werden, wenn die Sicherheitshinweise in den obengenannten Veröffentlichungen nicht verstanden werden.

Tradução das Informações Sobre Segurança

A legislação de muitos países obriga a que as informações sobre segurança, contidas em determianadas publicações, sejam apresentadas na língua do país onde o produto a que se referem é comercializado. O 6611 Network Processor Translated Safety Information, GA27-3954, contém as traduções das notas de **cuidado** e **perigo**, e outras informações sobre segurança que constam das publicações originais em língua inglesa:

- 6611 Network Processor Installation and Service Guide, GA27-3941
- 6611 Network Processor Operations Pocket Guide, GX27-3909.

Deverá consultar esta publicação sempre que não entender claramente qualquer informação sobre segurança constante das publicações inglesas.

Informações de Segurança Traduzidas

Em alguns países é obrigatório que as informações de segurança contidas nas publicações estejam em seus idiomas nacionais. A cartela *Informações de Segurança do 6611 Network Processor*, código GA17-0275-00, contém traduções dos avisos de cuidado e de perigo e outras informações que aparecem nas seguintes publicações em inglês:

- 6611 Network Processor Installation and Service Guide, GA27-3941
- 6611 Network Processor Operations Pocket Guide, GX27-3909.

Utilize a cartela acima quando não entender as informações de segurança existentes nessas publicações em inglês.

Información de Seguridad Traducida

Algunos países requieren que la información de seguridad contenida en las publicaciones se presente en su idioma nacional. El manual 6611 Network Processor Translated Safety Information, número de formulario GA27-3954, contiene traducciones de las notas de precaución y de peligro, y de otras informaciones de seguridad que aparecen en las siguientes publicaciones en inglés:

- 6611 Network Processor Installation and Service Guide, GA27-3941
- 6611 Network Processor Operations Pocket Guide, GX27-3909.

Consulte el manual siempre que no comprenda la información de seguridad de estas publicaciones en idioma inglés.

Oversatt sikkerhetsinformasjon

I noen land kreves det at sikkerhetsinformasjon i publikasjoner som ikke er oversatt, skal være oversatt til det respektive nasjonale språket. 6611 Network Processor Translated Safety Information, formnummer GA27-3954, inneholder en oversettelse av merknadene om advarsel og fare som forekommer i disse engelskspråklige håndbøkene:

- 6611 Network Processor Installation and Service Guide, GA27-3941
- 6611 Network Processor Operations Pocket Guide, GX27-3909.

Bruk dette heftet hvis du ikke forstår sikkerhetsinformasjonen i de engelskspråklige håndbøkene som er nevt ovenfor.

Suomenkieliset turvaohjeet

Julkaisu 6611 Network Processor Translated Safety Information, GA27-3954, sisältää seuraavissa julkaisuissa esiintyvien turvaohjeiden suomenkieliset käännökset:

- 6611 Network Processor Installation and Service Guide, GA27-3941
- 6611 Network Processor Operations Pocket Guide, GX27-3909.

Ennen kuin käsittelet IBM 6611 -verkkosuoritinta, lue toimeen liittyvät turvaohjeet. Jos et ymmärrä englanninkielistä turvaohjetta, lue vastaava suomenkielinen ohje.

Oversatte sikkerhedsforskrifter

Flere lande kræver, at sikkerhedsforskrifterne fra håndbøger og brugervejledninger skal foreligge på landets sprog. Håndbogen 6611 Network Processor Translated Safety Information, formnummer GA27-3954, indeholder oversættelser af sikkerhedsforskrifter mærket 'Pas på!' (CAUTION) og 'Fare!' (DANGER) og anden sikkerhedsinformation fra følgende håndbøger på engelsk:

- 6611 Network Processor Installation and Service Guide, GA27-3941
- 6611 Network Processor Operations Pocket Guide, GX27-3909.

Læs i vejledningen, når du ikke forstår sikkerhedsforskrifterne fra de engelske håndbøger.

Säkerhetsinformation

I vissa länder krävs att den säkerhetsinformation (varningstexter mm) som förekommer i handböcker ska finnas på resp lands språk. För Sverige gäller att sådan information i användarhandböcker ska finnas på svenska medan den får vara på engelska i böcker för servicepersonal. Dokumentet 6611 Network Processor Translated Safety Information, GA27-3954, innehåller översättningar av varningstexter och övrig säkerhetsinformation som förekommer i följande engelskspråkiga publikationer:

 6611 Network Processor Installation and Service Guide, GA27-3941 • 6611 Network Processor Operations Pocket Guide, GX27-3909.

Läs i dokumentet med den översatta säkerhetsinformationen om du inte förstår informationen i de engelska böckerna.

翻訳された安全に関する情報

国によっては、出版物に母国語で書かれた安全に関する情報を 記載することが義務づけられている場合があります。<u>IBM 66</u> <u>11 Network Processor Translated Safety Information</u>, GA27-3954 には、注意(CAUTION) および危険(DANGER) の但し 書き、また次の英語で書かれた出版物に記載されているその他 の安全に関する情報が含まれています。

- IBM 6611 Network Processor Installation and Service Guide, GA27-3941
- IBM 6611 Network Processor Operations Pocket
 <u>Guide,</u> GX27-3909.

これらの英語で書かれた出版物に記載されている安全に関する 情報がよくわからない場合は、いつでも小冊子を参照するよう にしてください。

안전 정보 번역서 몇몇 나라에서는 출×물에 포함된 안전 정보를 자국어로 표시하도록 규정하고 있 습니다. *IBM 6611 Network Processor Translated Safety Information*, GA27-3954 에는 다음의 영문× 서적에 나타나는 주의, 위험 통지사항과 / I타 안전 정보를 번역한 내용이 있습니다.

- IBM 6611 Network Processor Installation and Service Guide, GA27-3941
- IBM 6611 Network Processor Operations Pocket Guide, GX27-3909

이들 영문 서적에 있는 안전 정보나 이 해되지 않으면 언제든 이 책을 참조하십 시오.

Trademarks

The following terms, denoted by an asterisk (*) in this publication, are trademarks of the IBM Corporation in the United States or other countries or both:

AIX APPN IBM NetView Personal System/2 PS/2 RISC System/6000

The following terms, denoted by a double asterisk (**) in this publication, are trademarks of other companies as follows:

AppleTalk	Apple Computer, Inc.
DECnet	Digital Equipment
	Corporation
Hayes Smartmodem	Hayes Microcomputer
	Products, Inc.
XNS	Xerox Corporation

xxiv 6611 Operations Pocket Guide

About This Book

This book is organized in the following parts:

"Part I: IBM 6611 Network Processor Model 120"

Chapter 1 provides a description of all internal and external parts of the Model 120.

Chapter 2 provides information about operating the Model 120.

Chapter 3 provides information about relocating the Model 120.

"Part II: IBM 6611 Network Processor Model 140"

Chapter 4 provides a description of all internal and external parts of the Model 140.

Chapter 5 provides information about operating the Model 140.

Chapter 6 provides information about relocating the Model 140.

"Part III: IBM 6611 Network Processor Model 170"

Chapter 7 provides a description of all internal and external parts of the Model 170.

Chapter 8 provides information about operating the Model 170.

Chapter 9 provides information about relocating the Model 170.

"Part IV: Common Information for Models 120, 140, and 170"

Chapter 10 describes the attachment of an ASCII terminal to an IBM 6611.

Chapter 11 explains how to use a modem for remote IBM service.

Chapter 12 explains how to run the diagnostic programs.

Chapter 13 explains how to read the cable label.

Appendix A explains the operating and system initialization codes, and the 3-digit codes.

Appendix B explains how to order additional keys for the IBM 6611.

List of Abbreviations: Provides the definitions for abbreviated terms used in this book.

Who Should Use This Book

This book is intended for the customer. It provides basic information about how to operate the IBM 6611 Network Processor Models 120, 140, and 170.

Use the adhesive strip on the plastic pocket to attach it to the IBM 6611. Then, place this book inside the plastic pocket. This book should be kept with the IBM 6611 at all times.

Part I: IBM 6611 Network Processor Model 120

Chapter 1. Description of the IB	N		6	6-	11		M	00	je	ł		
120												1-1
Parts of the Model 120												1-3
External Device Connectors			•	•	•	•	•	•		•	•	1-5
Chapter 2. Using the Model 120												2-1
Setting the Key Mode Switch .												2-1
Starting Model 120 Operation .												2-3
Stopping Model 120 Operation												2-4
Reading the 3-Digit Display												2-4
Using the Reset Button												2-5
Loading a 3.5-Inch Diskette												2-6
Unloading a 3.5-Inch Diskette .		•	•	•	•	•	•	•	•	•		2-7
Chapter 3. Relocating the Mode	ł	1	2	0		•	•					3-1



Chapter 1. Description of the IBM 6611 Model 120

The IBM 6611 Network Processor Model 120 (*Model 120*) is designed to be placed on a tabletop or on a shelf in a standard Electronics Industries Association (EIA) 310-C rack.



Figure 1-1. Model 120

The Model 120 processes data from the disk drive, diskettes, and communication connections. It contains a 32-bit processor, network adapters, 16 MB of RAM, and media storage devices. The hardware is controlled by the IBM Multiprotocol Network Program, program number 5648-016.

The Model 120 has 2 communication adapters and a disk drive. Internal hardware controls a 3.5-inch diskette drive, 2 serial ports, and the disk drive.

The Model 120 does not have a keyboard or display screen. You can attach an ASCII terminal to the Model 120 to use the Multiprotocol Network Program System Manager functions, or you can use a workstation attached to an Internet Protocol (IP) network. The front operator panel of the Model 120 contains a 3-digit display, a Key Lock/Reset Button Assembly, and a 3.5-inch diskette drive.

© Copyright IBM Corp. 1992, 1993

The Multiprotocol Network Program is initially configured by first running the Multiprotocol Network Program Configuration Program on an IBM* Personal System/2* (PS/2*) or RISC System/6000* workstation and then transporting the configuration file to the Model 120 on a 3.5-inch diskette.

Subsequent configuration updates can be applied from a terminal or a RISC System/6000 workstation attached to the IP network.

Remote monitoring and control can be performed by communications with a Simple Network Management Protocol (SNMP) network manager station, such as the IBM NetView*/6000.

Parts of the Model 120

Front View



Rear View



Figure 1-2. Parts of the Model 120

The following list describes the parts of the Model 120, shown in Figure 1-2.

Covers can be removed to allow access to the inside of the Model 120.

Note: Only trained service technicians should work inside the Model 120.

Serial number is located on the front to the right of the 3-digit display.

Key mode switch is a key-controlled switch with three positions, labeled:

Normal Secure Service.

Note: The key mode switch also locks the cover of the Model 120.

Power switch has the international symbols I for On and **O** for Off. When the power switch is set to On, the power-on light-emitting diode (LED) comes on.

Operator panel door covers the 3-digit display, the in-use LED, and the diskette drive. It may be left open or closed.

Reset button is located next to the 3-digit display and is labeled **Reset**.

Diskette drive is a 1.44-MB, 3.5-inch diskette drive that has an in-use LED and a diskette-eject button.

3-digit display displays up to three 7-segment characters.

•

External Device Connectors

The connectors and ports located at the rear of the Model 120 are used to attach external devices.



Figure 1-3. Rear View of the Model 120

1-6 6611 Operations Pocket Guide

Chapter 2. Using the Model 120

This chapter describes the tasks associated with using the Model 120.

For information about connecting an ASCII terminal to the Model 120, see Chapter 10, "ASCII Terminal Installation and Service Tasks."

Setting the Key Mode Switch The key mode switch has three positions: Normal



Figure 2-1. Key Mode Switch Positions

Operation	Normal	Secure	Service			
Reset	Yes	No	Yes			
Normal IPL	Yes	No	No			
Service IPL	No	No	Yes			
Covers Locked	Yes	Yes	No			

Warning: The **Secure** position should not be used unless you are disconnecting the Model 120 from the network. The **Secure** position is used to prevent user intervention locally from the operator panel or remotely from the SNMP network manager.
The following section explains how to use the three positions on the key mode switch:

Warning: The Reset button is enabled when the key mode switch is in the **Normal** or **Service** position. If you press **Reset** and the Model 120 is still running, data can be damaged or lost.

- The Normal position is used for unattended operation. When the Model 120 IPLs, it proceeds according to the list of adapters established during the configuration of the Model 120.
- The Service position is used for attended operation when hardware or software service is performed. The Service position allows you to perform debug and dump procedures. In the Service position, the Model 120 attempts to IPL from the diskette drive, if it contains a diskette, or from the disk.
- The Secure position is used to prevent user intervention locally from the operator panel and remotely from the SNMP network manager. The Secure position should not be used unless the Model 120 is being disconnected from the network.

Before starting the Model 120 for operation, set the key mode switch to *Normal*. This permits the Model 120 software to load after the power-on self-test (POST) has completed.

If the Model 120 is not operating correctly, contact your IBM service representative. In the United States, call 1-800-IBM-SERV. In other countries, contact your IBM marketing representative for the service phone number.

Starting Model 120 Operation

To start the Model 120, perform the following procedure:

- 1. Set the key mode switch to Normal.
- Start the Model 120 by setting the power switch to On. The power-on LED will come on and the POST begins. During the POST, numbers are displayed on the 3-digit display.
- If the power-on LED does not come on, check the power cord, which is located at the rear of the Model 120, to ensure that it is plugged into a grounded electrical wall outlet. If the power cord is plugged in, contact your IBM service representative. In the United States, call 1-800-IBM-SERV. In other countries, contact your IBM marketing representative for the service phone number.



Figure 2-2. Front View of the Model 120

Stopping Model 120 Operation

To stop the Model 120, perform the following procedure:

Warning: Do not shut down the IBM 6611 without the knowledge and permission of the System Manager.

- 1. Stop the Model 120 software to prevent the loss of data. For information about how to stop the Model 120 software, refer to the *Multiprotocol Network Program User's Guide*.
- 2. Set the Model 120 power switch to Off.

Reading the 3-Digit Display

See "Reading the 3-Digit Display" on page A-1 for instructions about reading the error codes.

Using the Reset Button

Warning: When the key mode switch is in the **Normal** or **Service** position, pressing **Reset** causes the Model 120 to reset and start the system. If you press **Reset** and the Model 120 is still running, data can be damaged or lost.

The Reset button has two purposes:

- To cause an IPL of the Model 120 when the key mode switch is in the *Normal* or *Service* position.
- To display codes or diagnostic messages after a flashing 888 is displayed on the 3-digit display.



Figure 2-3. Location of the Reset Button

Loading a 3.5-Inch Diskette

To load a diskette into the drive, insert the diskette in the diskette drive with the labeled metal shutter first. Push the diskette into the drive until you hear a click. The click indicates that the diskette is securely positioned in the drive. The in-use LED is lit when the drive is being accessed.



Figure 2-4. Loading a Diskette

Unloading a 3.5-Inch Diskette

Warning: Do not stop the Model 120 or remove a diskette when the in-use LED is lit, or you may lose some of the data or damage the diskette.

"1.44" is printed on the diskette-eject button. To unload the diskette, push the diskette-eject button. The diskette ejects partially from the drive. Pull the diskette out.



Figure 2-5. Unloading a Diskette

2-8 6611 Operations Pocket Guide

Chapter 3. Relocating the Model 120

DANGER

An electrical outlet that is not correctly wired could place hazardous voltage on metal parts of the IBM 6611 or the devices that attach to the IBM 6611. It is the responsibility of the customer to ensure that the outlet is correctly wired and grounded to prevent an electrical shock.

Before installing or removing signal cables, ensure that the power cord for the IBM 6611 is unplugged.

When possible, use one hand to connect or disconnect signal cables to prevent a possible shock from touching two surfaces with different electrical potentials.

During an electrical storm, do not connect or disconnect any cables.

Perform the following tasks to prepare the Model 120 for relocation:

Warning: Do not shut down the IBM 6611 without the knowledge and permission of the System Manager.

- 1. Stop the Model 120 software to prevent the loss of data. For information about how to stop the Model 120 software, see the *Multiprotocol Network Program User's Guide*.
- 2. Set the power switch on the Model 120 to Off.
- 3. Label each cable connected to the rear of the Model 120. Then, disconnect each cable.
- 4. Set the key mode switch to *Secure* and remove the key. Tape the key to the Model 120.

5. Carefully package and relocate the Model 120 and associated material.

Warning: Check all of the power outlets in the location to which you are relocating the IBM 6611 for correct wiring, voltage, and grounding before connecting the IBM 6611 to the power outlets.

Part II: IBM 6611 Network Processor Model 140

Chapter 4. Description of the IBM 6611 Model	
140	4-1
Parts of the Model 140	4-3
External Device Connectors	4-5
Chapter 5. Using the Model 140	5-1
Key Mode Switch Positions	5-1
Starting Model 140 Operation	5-3
Stopping Model 140 Operation	5-4
Reading the 3-Digit Display	5-4
Using the Reset Button	5-5
Loading a 3.5-Inch Diskette	5-6
Unloading a 3.5-Inch Diskette	5-6
Changing the Pedestal on a Model 140	5-7
From Horizontal to Vertical Position	5-7
From Vertical to Horizontal Position	5-9
Changing the Position of the 3-Digit Display . 5	-13
Chapter 6 Belocating the Model 140	6-1



Chapter 4. Description of the IBM 6611 Model 140

The IBM 6611 Network Processor Model 140 (*Model* 140) is designed to be placed on a tabletop, in a vertical position on the floor, or on a shelf in a standard Electronics Industries Association (EIA) 310-C rack. The Model 140 has a removable pedestal that provides stability when the Model 140 is in the vertical position. The pedestal is attached to the bottom of the Model 140 when it is in the horizontal position.



Figure 4-1. Example of Vertical and Horizontal Positions for the Model 140

The Model 140 processes data from the disk drives, diskettes, and communication connections. It contains a 32-bit processor, network adapters, 16 MB of RAM, and media storage devices. The hardware is controlled by the IBM Multiprotocol Network Program, program number 5648-016.

The Model 140 allows up to 4 communication adapters to be installed and has two 160-MB disk drives. Internal hardware controls a 3.5-inch diskette drive, 2 serial ports, and the disk drives.

The Model 140 does not have a keyboard or display screen. You can attach an ASCII terminal to the Model 140 to use the Multiprotocol Network Program System Manager functions, or you can use a workstation attached to an Internet Protocol (IP) network. The front operator panel of the Model 140 contains a 3-digit display, a Key Lock/Reset Button Assembly, and a 3.5-inch diskette drive.

The Multiprotocol Network Program is initially configured by first running the Multiprotocol Network Program Configuration Program on an IBM Personal System/2 (PS/2) or RISC System/6000 workstation and then transporting the configuration file to the Model 140 on a 3.5-inch diskette.

Subsequent configuration updates can be applied from a terminal or a RISC System/6000 workstation attached to the IP network.

Remote monitoring and control can be performed by communications with a Simple Network Management Protocol (SNMP) network manager station, such as the IBM NetView/6000.

Parts of the Model 140



Figure 4-2. Model 140 with Rear Cover Removed

The following list describes the parts of the Model 140, shown in Figure 4-2 on page 4-3.

Covers can be removed to allow access to the rear of the Model 140. If the Model 140 is to be operated in a wiring closet, it is desirable to operate it without the rear cover for purposes of cooling.

Note: Only trained service technicians should work inside the Model 140.

Serial number is located in the left front corner of the top cover.

Key mode switch is a key-controlled switch with three positions, labeled:

Normal Secure Service.

Note: The key mode switch also locks the cover of the Model 140.

Power switch has the international symbols I for On and **O** for Off. When the power switch is set to On, the power-on light-emitting diode (LED) comes on.

Reset button is located next to the 3-digit display and is labeled **Reset**.

Diskette drive is a 1.44-MB, 3.5-inch diskette drive that has an in-use LED and a diskette-eject button.

Power supply fan is located in the rear left corner of the Model 140.

Rear fan is located in the right rear corner of the Model 140.

3-digit display displays up to three 7-segment characters. The display has a hinged lens that allows access to the character holder, which can be rotated 90 degrees for correct viewing when the Model 140 is in the horizontal or vertical position.

3-Digit Display



Hinged Lens (Open)

Figure 4-3. Front View of the Model 140 with the 3-Digit Display Hinged Lens Open

Pedestal is removable and can be attached to the left side or the bottom to place the Model 140 in the vertical or horizontal position, respectively.

External Device Connectors

When the rear cover is removed, the Model 140 connectors and ports are accessible. These connectors and ports are used to attach external devices.



Figure 4-4. Rear of the Model 140 Serial Ports and Adapter Slots

4-6 6611 Operations Pocket Guide

Chapter 5. Using the Model 140

This chapter describes the tasks associated with using the Model 140.

For information about connecting an ASCII terminal to the Model 140, see Chapter 10, "ASCII Terminal Installation and Service Tasks."

Key Mode Switch Positions

The key mode switch has three positions:

Normal Secure Service.



Key Mode Switch

Figure 5-1. Key Mode Switch Positions

Operation	Normal	Secure	Service
Reset	Yes	No	Yes
Normal IPL	Yes	No	No
Service IPL	No	No	Yes
Covers Locked	Yes	Yes	No

Warning: The **Secure** position should not be used unless you are disconnecting the Model 140 from the network. The **Secure** position is used to prevent user intervention locally from the operator panel or remotely from the SNMP network manager. The following section explains how to use the three positions on the key mode switch:

Warning: The Reset button is enabled when the key mode switch is in the **Normal** or **Service** position. If you press **Reset**, data can be damaged or lost if the Model 140 is still running.

- The *Normal* position is used for unattended operation. When the Model 140 IPLs, it proceeds according to the list of adapters established during the configuration of the Model 140.
- The Service position is used for attended operation when hardware or software service is performed. The Service position allows debug and dump procedures. In the Service position, the Model 140 attempts to IPL from the diskette drive, if it contains a diskette, or from the disk.
- The Secure position is used to prevent user intervention locally from the operator panel and remotely from the SNMP network manager. The Secure position should not be used unless you are disconnecting the Model 140 from the network.

Before starting the Model 140 for operation, set the key mode switch to *Normal*. This permits the Model 140 software to load after the power-on self-test (POST) has completed.

If the Model 140 is not operating correctly, contact your IBM service representative. In the United States, call 1-800-IBM-SERV. In other countries, contact your IBM marketing representative for the service phone number.

Starting Model 140 Operation

To start the Model 140, perform the following procedure:

- 1. Set the key mode switch to Normal.
- Start the Model 140 by setting the power switch to On. The power-on LED will come on and the POST begins. During the POST, numbers are displayed on the 3-digit display.
- If the power-on LED does not come on, check the power cord, which is located at the rear of the Model 140, to ensure that it is plugged into a grounded electrical wall outlet. If the power cord is plugged in, contact your IBM service representative. In the United States, call 1-800-IBM-SERV. In other countries, contact your IBM marketing representative for the service phone number.



Figure 5-2. Front View of the Model 140

Stopping Model 140 Operation

To stop the Model 140, perform the following procedure:

Warning: Do not shut down the IBM 6611 without the knowledge and permission of the System Manager.

- 1. Stop the Model 140 software to prevent the loss of data. For information about how to stop the Model 140 software, refer to the *Multiprotocol Network Program User's Guide*.
- 2. Set the Model 140 power switch to Off.

Reading the 3-Digit Display

See "Reading the 3-Digit Display" on page A-1 for instructions about reading the error codes.

Using the Reset Button

Warning: When the key mode switch is in the **Normal** or **Service** position, pressing **Reset** causes the Model 140 to reset and start the system. If you press **Reset**, data can be damaged or lost if the Model 140 is still running.

The Reset button has two purposes:

- To cause an IPL of the Model 140 when the key mode switch is in the *Normal* or *Service* position.
- To display codes or diagnostic messages after a flashing 888 is displayed on the 3-digit display.

Reset Button



Figure 5-3. Location of the Reset Button

Loading a 3.5-Inch Diskette

To load a diskette into the drive, insert the diskette in the diskette drive with the labeled metal shutter first. Push the diskette into the drive until you hear a click. The click indicates that the diskette is securely positioned in the drive. The in-use LED is on when the drive is being accessed.



Figure 5-4. Loading a Diskette

Unloading a 3.5-Inch Diskette

Warning: Do not stop the Model 140 or remove a diskette when the in-use LED is on, or you may lose some of the data or damage the diskette.

"1.44" is printed on the diskette-eject button. To unload the diskette, push the diskette-eject button. The diskette ejects partially from the drive. Pull the diskette out.



Figure 5-5. Unloading a Diskette

Changing the Pedestal on a Model 140

The Model 140 operates in either the horizontal or vertical position. The vertical position is recommended when the Model 140 is placed on the floor.

When the Model 140 is used in the horizontal position, the pedestal is attached to the bottom of the unit with 2 large screws. When the Model 140 is used in the vertical position, the pedestal can be removed from the bottom and attached to the side that has 2 screw-mounting holes. The screws have knobs for ease in attaching or removing the pedestal. No screwdriver is required.

From Horizontal to Vertical Position

To change the unit from the horizontal to the vertical position:

- See "Stopping Model 140 Operation" on page 5-4 before changing the Model 140 pedestal.
- 2. Set the Model 140 power switch to Off and unplug the power cord from the outlet.
- Label each cable connected to the rear of the Model 140. Then, disconnect each cable.
- Stand the Model 140 on its side with the 2 screw-mounting holes on the left side facing upward.
- 5. Remove the 2 screws that secure the pedestal to the Model 140.
- 6. Remove the pedestal.

Warning: In Figure 5-6 on page 5-8, the Model 140 is shown placed on the side that does not have screw-mounting holes. *Do not* operate the Model 140 when it is in this position. This may result in damage to the disk drive. Place the pedestal on the side of the Model 140 and align it with the screw-mounting holes.
Place the pedestal so that the part number imprinted on the bottom of the pedestal is toward the front of the Model 140.



Figure 5-6. Attaching the Pedestal for the Vertical Position

8. Finger-tighten the 2 screws on the Model 140.

9. Stand the Model 140 in the vertical position.



- See "Changing the Position of the 3-Digit Display" on page 5-13 for instructions to adjust the 3-digit display.
- 11. Reconnect any cables that were connected to the rear of the Model 140.
- 12. Plug the power cord into the outlet.

From Vertical to Horizontal Position

To change the Model 140 from the vertical to the horizontal position:

- 1. See "Stopping Model 140 Operation" on page 5-4 before changing the Model 140 pedestal.
- 2. Set the Model 140 power switch to Off and unplug the power cord from the outlet.
- Ensure that all cables connected to the rear of the Model 140 are labeled as you disconnect them.

 Tilt the Model 140 downward and rest it with the top facing upward and the bottom facing downward.



Figure 5-7. Removing the Pedestal in the Vertical Position

5. Remove the 2 screws that secure the pedestal to the Model 140.



Figure 5-8. Attaching the Pedestal for the Horizontal Position

- 7. Stand the Model 140 on its right side with the 2 screw-mounting holes located on the left side facing upward.
- 8. Place the pedestal on the bottom of the Model 140 and align it with the screw-mounting holes.

- 9. Finger-tighten the 2 screws on the Model 140.
- 10. Stand the Model 140 in the horizontal position on the pedestal.
- See "Changing the Position of the 3-Digit Display" on page 5-13 for instructions to adjust the 3-digit display.
- 12. Reconnect any cables that were connected to the rear of the Model 140.
- 13. Plug the power cord into the outlet.

Changing the Position of the 3-Digit Display

You can change the position of the 3-digit display so that it can be read easily when the Model 140 is in either the vertical or horizontal position.

Changing the 3-Digit Display from Vertical to Horizontal Position: To change the 3-digit display from the vertical position to the horizontal position (with the Model 140 in the horizontal position), perform the following procedure:

Note: In the following figures, the three 7-segment characters are lit to show the position of the display.



Figure 5-9. 3-Digit Display in the Vertical Position with the Hinged Lens Closed

1. Open the hinged lens.



Figure 5-10. 3-Digit Display in the Vertical Position with the Hinged Lens Open

2. Hold the character holder with your fingertips.

3. Turn the character holder 90 degrees clockwise.



Figure 5-11. 3-Digit Display in the Horizontal Position with the Hinged Lens Open

4. Close the hinged lens.



Figure 5-12. 3-Digit Display in the Horizontal Position with the Hinged Lens Closed

Changing the 3-Digit Display from Vertical to Horizontal Position: To change the 3-digit display from the vertical position to the horizontal position (with the Model 140 in the vertical position), perform the following procedure:

Note: In the following figures, the three 7-segment characters are lit to show the position of the display.

1. Open the hinged lens.



Figure 5-13. 3-Digit Display in the Vertical Position with the Hinged Lens Open

2. Hold the character holder with your fingertips.

3. Turn the character holder 90 degrees counterclockwise.



Figure 5-14. 3-Digit Display in the Horizontal Position with the Hinged Lens Open

4. Close the hinged lens.



Figure 5-15. 3-Digit Display in the Horizontal Position with the Hinged Lens Closed

5-18 6611 Operations Pocket Guide

Chapter 6. Relocating the Model 140

An IBM service representative should be contacted to disconnect and relocate your Model 140. In the United States, call 1-800-IBM-SERV. In other countries, contact your IBM marketing representative for the service phone number.

Warning: Damage as a result of incorrect handling may void your equipment warranty. Contact your local IBM service representative to obtain assistance when you prepare your Model 140 for moving.

Do not shut down the IBM 6611 without the knowledge and permission of the System Manager.

CAUTION:

Check all of the power outlets in the location to which you are relocating the IBM 6611 for correct wiring, voltage, and grounding before allowing the IBM Service Representative to connect the IBM 6611 to the power outlets.
6-2 6611 Operations Pocket Guide

.

Part III: IBM 6611 Network Processor Model 170

Chapter 7. Description of the IBM 6611 Model	
170	7-1
Parts of the Model 170	7-3
External Device Connectors	7-6
Chapter 8. Using the Model 170	8-1
Setting the Key Mode Switch	8-1
Starting Model 170 Operation	8-4
Stopping Model 170 Operation	8-5
Reading the 3-Digit Display	8-5
Using the Reset Button	8-6
Loading a 3.5-Inch Diskette	8-7
Unloading a 3.5-Inch Diskette	8-8
Chapter 9. Relocating the Model 170	9-1



Chapter 7. Description of the IBM 6611 Model 170

The IBM 6611 Network Processor Model 170 (*Model* 170) is designed to be placed on the floor. The Model 170 contains a 32-bit processor, a small computer systems interface (SCSI) adapter, 16 MB of RAM, a 3.5-inch diskette drive, a 355-MB disk drive, and network adapters.

The Model 170 does not have a keyboard or display screen. You can attach an ASCII terminal to the Model 170 to use the Multiprotocol Network Program System Manager functions, or you can use a workstation attached to an Internet Protocol (IP) network. The front operator panel of the Model 170 contains a 3-digit display, a Key Lock/Reset Button Assembly, and a 3.5-inch diskette drive.

The Multiprotocol Network Program is initially configured by first running the Multiprotocol Network Program Configuration Program on an IBM Personal System/2 (PS/2) or RISC System/6000 workstation and then transporting the configuration file to the Model 170 on a 3.5-inch diskette.

Subsequent configuration updates can be applied from a terminal or a RISC System/6000 workstation attached to the IP network.

Remote monitoring and control can be performed via communications with a SNMP network manager station, such as the IBM NetView/6000 workstation.

The Model 170 has 8 adapter slots. Seven are for communication adapters, and the eighth slot contains the standard SCSI adapter.



Figure 7-1. Model 170

Parts of the Model 170

Front Cover



Figure 7-2. Front View of the Model 170



Figure 7-3. Side View of the Model 170



Figure 7-4. Rear View of the Model 170

The following list describes the parts of the Model 170, shown in Figure 7-2 on page 7-3 through Figure 7-4 on page 7-4.

Note: Only trained service technicians should work inside the Model 170.

Serial number is located below the power switch.

Key mode switch is a key-controlled switch with three positions, labeled:

Normal Secure Service.

Power switch has the international symbols I for On and **O** for Off. When the power switch is set to On, the power-on LED comes on.

Reset button is located next to the 3-digit display and is labeled **Reset**.

Diskette drive is a 1.44-MB, 3.5-inch diskette drive that has an in-use LED and a diskette-eject button.

3-digit display displays up to three 7-segment characters.

Front and side covers are removable and allow access to the inside of the Model 170. **Rear cover** is removable and provides access to the Model 170 connectors.

Note: Only trained service technicians should work inside the Model 170.

Rear cover lock provides physical security by preventing the removal of any covers when locked. The key mode switch and rear cover lock use the same key. The rear cover lock is locked when the key slot points to the locked symbol and is unlocked when the key slot is turned clockwise to the open symbol. **Hand slot** is located on the bottom center of the front cover and can be used to raise the front of the Model 170 when positioning it.

Cable guide allows orderly arrangement of the cables.

Rollers are designed for easy placement of the Model 170.

External Device Connectors

When the rear cover is removed, the Model 170 connectors and ports are accessible. These connectors and ports are used to attach external devices. Only trained service technicians should remove the rear cover.

Chapter 8. Using the Model 170

This chapter describes the tasks associated with using the Model 170.

For information about connecting an ASCII terminal to the Model 170, see Chapter 10, "ASCII Terminal Installation and Service Tasks."

Setting the Key Mode Switch

The key mode switch has three positions:

Normal Secure Service.



Figure 8-1. Key Mode Switch Positions

Operation	Normal	Secure	Service
Reset	Yes	No	Yes
Normal IPL	Yes	No	No
Service IPL	No	No	Yes
Covers Locked	Yes	Yes	No

Warning: The Secure position should not be used unless you are disconnecting the Model 170 from the network. The Secure position is used to prevent user intervention locally from the operator panel or remotely from the SNMP network manager.

The following section explains how to use the three positions on the key mode switch:

Warning: The Reset button is enabled when the key mode switch is in the **Normal** or **Service** position. If you press **Reset**, data can be damaged or lost if the Model 170 is still running.

- The *Normal* position is used for unattended operation. The IPL proceeds according to the list of adapters established during the configuration of the Model 170.
- The Service position is used for attended operation when hardware or software service is performed. The Service position allows debug and dump procedures. In the Service position, the Model 170 attempts to IPL from the diskette drive, if it contains a diskette, or from the disk.
- The Secure position is used to prevent user intervention locally from the operator panel and remotely from the SNMP network manager. The Secure position should not be used unless you are disconnecting the Model 170 from the network.

Before starting the Model 170 for operation, set the key mode switch to *Normal*. This permits the Model 170 software to load after the POST has completed.

If the Model 170 is not operating correctly, contact your IBM service representative. In the United States, call 1-800-IBM-SERV. In other countries, contact your IBM marketing representative for the service phone number.

Starting Model 170 Operation

To start the Model 170, perform the following procedure:

- 1. Set the key mode switch to Normal.
- Start the Model 170 by setting the power switch to On. The power-on LED will come on and the POST begins. During the POST, numbers are displayed on the 3-digit display.
- If the power-on LED does not come on, check the power cord, which is located at the rear of the Model 170, to ensure that it is plugged into a grounded electrical wall outlet. If the power cord is plugged in, contact your IBM service representative. In the United States, call 1-800-IBM-SERV. In other countries, contact your IBM marketing representative for the service phone number.



Figure 8-2. Front View of the Model 170

Stopping Model 170 Operation

To stop the Model 170, perform the following procedure:

Warning: Do not shut down the IBM 6611 without the knowledge and permission of the System Manager.

- 1. Stop the Model 170 software to prevent the loss of data. For information about how to stop the Model 170 software, refer to the *Multiprotocol Network Program User's Guide*.
- 2. Set the Model 170 power switch to Off.

Reading the 3-Digit Display

See "Reading the 3-Digit Display" on page A-1 for instructions to read the error codes.

Using the Reset Button

Warning: The Reset button is enabled when the key mode switch is in the Normal or Service position. If you press **Reset**, data can be damaged or lost if the Model 170 is still running.

The Reset button has two purposes:

- To cause a system start of the Model 170 when the key mode switch is in the *Normal* or *Service* position.
- To display codes or diagnostic messages after a flashing 888 is displayed on the 3-digit display.



Figure 8-3. Location of the Reset Button

Loading a 3.5-Inch Diskette

To load a diskette into the drive, insert the diskette in the diskette drive, with the labeled metal shutter first. Push the diskette into the drive until you hear a click. The click indicates that the diskette is securely positioned in the drive. The in-use LED is on when the drive is being accessed.



Figure 8-4. Loading a Diskette

Unloading a 3.5-Inch Diskette

Warning: Do not stop the Model 170 or remove a diskette when the in-use LED is on, or you may lose some of the data or damage the diskette.

"1.44" is printed on the diskette-eject button. To unload the diskette, push the diskette-eject button. The diskette ejects partially from the drive. Pull the diskette out.



Figure 8-5. Unloading a Diskette

Chapter 9. Relocating the Model 170

An IBM service representative should be contacted to disconnect and relocate your Model 170. In the United States, call 1-800-IBM-SERV. In other countries, contact your IBM marketing representative for the service phone number.

Warning: Damage as a result of incorrect handling may void your equipment warranty. Contact your local IBM service representative for assistance to prepare your Model 170 for moving.

Do not shut down the IBM 6611 without the knowledge and permission of the System Manager.

CAUTION:

Check all of the power outlets in the location to which you are relocating the IBM 6611 for correct wiring, voltage, and grounding before allowing the IBM Service Representative to connect the IBM 6611 to the power outlets.

9-2 6611 Operations Pocket Guide

Part IV: Common Information Models 120, 140, and	for 170
Chapter 10. ASCII Terminal Installation and	
Service Tasks	10-1
General Setup Attributes	10-2
Communication Setup Attributes	10-3
Terminal Cable	10-6
ASCII Terminal Cable	10-8
Chapter 11. Using a Modem for Remote IBM	
Service	11-1
Chapter 12. Problem Determination	12-1
The Functions of the Diagnostic Program	12-1
Diagnostic Routines	12-1
Running Concurrent Mode Diagnostics	12-1
Running Concurrent Mode Diagnostics on a	
Direct-Attached ASCII terminal	12-2
Running Concurrent Mode Diagnostics on a	
Modem-Attached ASCII terminal	12-3
Running Concurrent Mode Diagnostics via	
the IP Network	12-3
Service Aids	12-4
Service Hints	12-4
Display Previous Diagnostic Results	12-4
Display or Change Configuration or Vital	
Product Data (VPD)	12-5
Display Machine Configuration	12-5
Display VPD	12-5
Display or Alter VPD	12-5
Display or Change Diagnostic Test List	12-5
Diskette Media	12-5
Reset Bootlist	12-6
Chapter 13. Cable Label	13-1
Understanding the Cable Label	13-1
Explanation of Label Fields	13-3
IBM 6611 Unit Number or Name	13-3
IBM 6611 Slot Number	13-3
Adapter Type	13-4

© Copyright IBM Corp. 1992, 1993

Adapter Cable Type	13-5
This Cable Connects to Device Type	13-5
This Cable Connects to Device Location	13-5
This Cable Connects to Device Number	
or Name	13-5
Network Number or Name	13-6
IBM 6611 ID at Network Manager	13-6
IBM 6611 Address at Network	13-6
Network Manager Location	13-6
Network Manager Operator's Phone	
Number	13-6

Chapter 10. ASCII Terminal Installation and Service Tasks

An ASCII terminal can be connected to one of the serial ports built into the rear of the IBM 6611. The serial ports are labeled *Serial 1* and *Serial 2* on the Model 140, and *S1* and *S2* on the Models 120 and 170.

If Serial 2 or S2 has been configured to support network management of a Cylink 4201 Advanced Channel Service Unit (ACSU), it cannot be used for connection of an ASCII terminal.

All models of the IBM 6611 support the following ASCII terminals as well as devices, such as personal computers, that are configured to emulate one of these terminals:

- IBM terminals
 - IBM 3101
 - IBM 3151
 - IBM 3161
 - IBM 3162
 - IBM 3163
 - IBM 3164
 - IBM AIX* TERMinal (a terminal emulation program for the RISC System/6000 computer)
- DEC terminals
 - VT100
 - VT320
 - VT330
 - VT340.

Note: If you use a personal computer that is configured to emulate one of the supported ASCII terminals, the display must support a 24 x 80 panel format (24 rows of 80 characters).

General Setup Attributes: Use Table 10-1 on page 10-2 to determine the settings for the general setup attributes for ASCII terminals or emulators.

Note: If you want to use an ASCII terminal or emulator that does not emulate one of the IBM ASCII terminals, refer to the operator's manual for that device, compare its attributes with those in the table, and set them accordingly.

General Setup		
Attributes	Settings	Explanations of the Settings
Row and Column	24 x 80	The screen format is 24 rows of 80 characters.
Scroll	Jump	When the last character on the last line of the screen is received or entered, all of the lines on the screen move up one line.
Auto LF	Off	For the Off setting, pressing Return moves the cursor to the first character position of the current line. The carriage return (CR) and the line feed (LF) characters are generated by the setting for the New Line attribute.
CRT saver	Off	The Off setting disables screen blanking.
Line wrap	On	The cursor moves to the first character position of the next line on the screen after it reaches the last character position of the current line on the screen.
Forcing insert	Off	If there is no space to insert a character or a null line, an insert operation cannot be performed.

 Table 10-1 (Page 1 of 2). Communication Setup

 Attributes for the ASCII Terminal or Emulator

General Setup	Cottingo	Evelopetions of the Collinse
Allfibules	Settings	Explanations of the Settings
Tab	Field	The column tab stops are ignored, and tab operation depends on the field attribute character positions.
Trace	All	Both inbound data (data to the IBM 6611) and outbound data (data from the IBM 6611) to and from the main port can be transferred to the terminal's auxiliary port without disturbing communications with the IBM 6611 when Trace is pressed.

 Table 10-1 (Page 2 of 2). Communication Setup

 Attributes for the ASCII Terminal or Emulator

Communication Setup Attributes: The settings for the attributes in Table 10-2 on page 10-4 are based on the assumption that the ASCII terminal or emulator is directly connected to the serial port of the IBM 6611. If the ASCII terminal communicates with the IBM 6611 via a modem connected to the serial port, a different setting for some attributes may be required. A dagger (†) indicates those attributes that may require different settings. The appropriate setting depends on the type of modem and the communication line. Not all functions exist in all terminal models.

Communication Setup Attributes	Settings	Explanations of the Settings
Operating mode	Echo	Data entered from the keyboard on the terminal is sent to the system unit for translation and then back to the display panel. Echo mode is also called <i>conversation mode</i> .
Line speed†	2400, 9600, 19200, 38400	Uses 9600-bps, 19200-bps, or 38400-bps line speeds to communicate with the IBM 6611 for a direct-attached terminal. Uses 2400-bps line speed to communicate with the IBM 6611 for a modem-attached terminal.
Word length	8	The data word length is 8 bits.
Parity	No	Does not add a parity bit, and is used together with the word length attribute to form the 8-bit data word (byte).
Stop bit	1	Places 1 bit after a data word.
Turnaround character	CR	The carriage return (CR) character is used as the line turnaround character.
Interface	RS232C	Uses the EIA 232-C electrical interface protocol.
Line control†	IPRTS	Uses the permanent request to send (IPRTS) signal to communicate with the IBM 6611.

Table 10-2 (Page 1 of 2). Communication SetupAttributes for the ASCII Terminal or Emulator

Communication Explanations Setup Attributes Settinas of the Settings Break signalt 500 The terminal sends a 500-ms break signal to the IBM 6611 after the Break key is pressed. Send null On The terminal does not suppress (if send trailing null characters to the IBM supported) 6611. Send null (if On The terminal sends trailing null characters to supported) the IBM 6611. Response 100 The terminal waits 100 ms delay† for the IBM 6611 to respond. Enter Return The Enter key functions as the Return key. Return New The cursor moves to the line next unprotected field when the Return key is pressed. New line CR The Return key generates the carriage return (CR) and the line feed (LF) characters. The line turnaround occurs after the CR and LF characters have been generated. Send Page The contents of the current page are sent to the IBM 6611 when the Send key is pressed. Insert character Space A blank character is inserted when the Insert key is pressed.

 Table
 10-2
 (Page 2 of 2).
 Communication Setup

 Attributes for the ASCII Terminal or Emulator
 Image: Setup 2 of 2).
 Image: Setup 2 of 2).
 Image: Setup 2 of 2).

Terminal Cable

Model 120: You must provide a cable to connect the ASCII terminal or emulator to the EIA 232 serial port. A terminal interposer is shipped with the Model 120. You need the terminal interposer to connect the terminal cable to the IBM 6611 as illustrated in Table 10-3 and Figure 10-1.

Index	Name	Description
A	EIA 232 serial port	25-pin D, male
В	Terminal cable connector, IBM 6611 end	25-pin D, female
С	Terminal cable	Customer supplied
D	Terminal interposer	Shipped with the IBM 6611

 Table 10-3. ASCII Terminal Connection to an EIA 232 Serial

 Port for Model 120

Note: A gender matcher may be required to connect your terminal to the terminal interposer.



Figure 10-1. Connecting a Terminal to an EIA 232 Serial Port for Model 120

Models 140 and 170: You must provide a cable to connect the ASCII terminal or emulator to the EIA 232 serial port. Two 10-to-25-pin converter cables and a terminal interposer are shipped with the IBM 6611. You need one of these converter cables and the terminal interposer to connect the terminal cable to the IBM 6611, as illustrated in Table 10-4 on page 10-7 and Figure 10-2 on page 10-7.

Index	Name	Description		
A	EIA 232 serial port	EIA 232-D, 10-pin square, male		
B	Converter cable connector, IBM 6611 end	10-pin square, female		
C	Converter cable connector, ASCII terminal end	25-pin D, male		
D	10-to-25-pin converter cable	Shipped with the IBM 6611		
E	Terminal cable connector, IBM 6611 end	25-pin D, female		
F	Terminal cable 25-pin D, m connector, terminal end			
G	Terminal cable	Customer supplied		
н	Terminal interposer	Shipped with the IBM 6611		

 Table
 10-4.
 ASCII Terminal Connection to an EIA 232 Serial

 Port for Models
 140 and 170

Note: A gender matcher may be required to connect your terminal to the terminal interposer.



Figure 10-2. Connecting a Terminal to an EIA 232 Serial Port for Models 140 and 170

The ASCII terminal cable you provide must be compatible with the 25-pin D connector on the terminal interposer. Table 10-5 on page 10-8 describes how the pins on both ends of the terminal interposer are wired, and Figure 10-3 on page 10-8 illustrates both ends.

IBM 6611 End Pin Number	Terminal End Pin Number	
3	2	
2	3	
6	20	
8	20	
20	8	
20	6	
5	4	
4	5	
7	7	
1	Shield	

Table 10-5. Terminal Interposer Pin Wiring



Figure 10-3. Terminal Interposer Connectors

ASCII Terminal Cable: The Model 140 and Model 170 are shipped with a 10-to-25-pin converter cable that is used to connect the ASCII terminal cable to the serial port.

Table 10-6 on page 10-9 and Figure 10-4 on page 10-9 describe the connectors on both ends of the 10-to-25-pin converter cable. Use the description of the ASCII terminal end of the cable to help identify the type of ASCII terminal cable you will need to connect the ASCII terminal to the converter cable.

Table	10-6.	10-to-25	Pin	Converter	Cable	Pin
Assigr	nment	s				

Port End Pin Number	Terminal End Pin Number	Signal
1	2	Transmit Data
2	20	Data Terminal Ready
3	4	Request to Send
4	22	Ring Indicate
5		Reserved
6	3	Receive Data
7	6	Data Set Ready
8	5	Clear to Send
9	8	Data Carrier Detect
10	7	Signal Ground
5	10	13 25
6611 E	nd	ASCII Terminal

Figure 10-4. 10-to-25-Pin Converter Cable Connectors

10-10 6611 Operations Pocket Guide

Chapter 11. Using a Modem for Remote IBM Service

In certain situations when you call for IBM service, the IBM service representative will request permission to establish a temporary, remote connection to the IBM 6611. A modem must be connected to the serial port. It must be a 2400-bps modem that is compatible with the Hayes Smartmodem** Attention (AT) command set.

Whenever you authorize IBM service to establish such a remote connection, you must:

- Connect and configure the modem before the remote service session begins. (The System Manager provides a function that enables you to configure the modem remotely.)
- N

For an illustration of the connection of a remote modem, see the information about remote access using a modem in the *IBM* 6611 *Network Processor Introduction and Planning Guide*.

- Provide the IBM service representative with the following information:
 - The telephone number of a modem that is (or will be) connected to the S2 or Serial 2 port of the IBM 6611.
 - A password that is valid only during the time that the problem exists. The IBM service representative must use the password to log in to the IBM 6611. You are responsible for deleting the password authorization after the problem has been solved.

11-2 6611 Operations Pocket Guide

Chapter 12. Problem Determination

The Functions of the Diagnostic Program

Diagnostic Routines: Diagnostic Routines serve to test the system hardware and detect hardware problems. Problems are indicated by Service Request Numbers (SRNs). An SRN allows the service representative to determine quickly what parts are required to repair the system.

Diagnostics can be run in concurrent mode, while the system is operating, or in stand-alone mode, when the system operation has been shut down.

Service Aids: The diagnostic package contains service aids. These service aids are used to display data and perform additional testing.

Diagnostic Routines

Warning: Do not shut down the IBM 6611 without the knowledge and permission of the System Manager.

Running Concurrent Mode Diagnostics

Concurrent mode diagnostics are diagnostics that can be run while the system is operating. They are designed to test as much of the system, or specified component, as possible without interfering with system operations. To accomplish this, it is necessary to limit diagnostic testing. If the resource cannot be tested totally in this mode because the resource is in use by the system, a message will be displayed informing the operator. The operator then has the option of stopping some of the system in order to free the resource for more complete testing. The operator can use the facilities of the System Manager to perform an orderly shutdown of the system and test it in stand-alone mode. Running concurrent diagnostics requires either an ASCII terminal attached directly or via a modem to the serial port, or a remote workstation attached to the Internet Protocol (IP) network.

Running Concurrent Mode Diagnostics on a Direct-Attached ASCII terminal

An ASCII terminal attached to the serial port of an IBM 6611 network controller will be active as a workstation. Follow the instructions specified in the *Multiprotocol Network Program User's Guide*, for logging in. At the initial panel, System Management Operating Instructions, select **Concurrent Hardware Diagnostics.** When the next menu, Diagnostic Operating Instructions, appears, press **Enter**.

At the next menu, Function Selection, select **Diagnostic Routines** by moving the cursor to that line and pressing **Enter**.

The next menu, Diagnostic Mode Selection, will give the operator the option of selecting System Verification or Problem Determination. The latter will, in addition to running the hardware diagnostics, analyze the error log to determine whether system errors have been detected during the past 24 hours. Select the mode of operation by moving the cursor using the up-arrow or down-arrow key to the desired selection and then pressing **Enter**. Select **System Verification** to verify correct operation of the system after repairs have been completed. Select **Problem Determination** to test the system when you suspect a problem.

Whichever mode you select, the next menu will provide a list of the resources that can be tested. Select the resource to be tested by moving the cursor up or down using the up-arrow or the down-arrow key, and press **Enter** when the cursor is at the correct position. To exit from any menu to the previous menu, press **F3**. This key may not be the same on all terminals, or it may not have the same key code. If there is no **F3** key or if it does not work, press **Escape** followed immediately by **3**.

Running Concurrent Mode Diagnostics on a Modem-Attached ASCII terminal

If the system to be tested has a modem connected to its serial port, tests can be run remotely. The telephone number, baud rate, password, and settings must be known. For more information, see Table 10-2 on page 10-4. With a terminal or emulator and a modem set to the same baud rate and settings as the modem attached to the IBM 6611, use the instructions for your modem and terminal to dial the IBM 6611 as you would a remote host. When the connection is completed, operation will be the same as for a direct-attached ASCII terminal.

Running Concurrent Mode Diagnostics via the IP Network

At a workstation that has access to the same IP network as the IBM 6611, enter the command **tn xxx.xxx.xxx** where **xxx.xxx.xxx** is the IP address of the IBM 6611 to be accessed. If the user has allowed remote access to this IBM 6611, the login panel for the IBM 6611 should be displayed. From then on, operation is the same as for a direct-attached ASCII terminal.
Service Aids

Service aids are utility programs provided for the convenience of the customer. The following service aids are provided:

- Service Hints
- Display Previous Diagnostic Results
- Display or Change Configuration or Vital Product Data (VPD)
- Display or Alter VPD
- Display or Change Diagnostic Test List
- Diskette Media
- Reset Bootlist.

Service Hints

This selection displays service hints and errata information about the maintenance package. It also contains information about using this particular version of the diagnostic package. Use the **PgDn** key to page forward through the information, or the **PgUp** key to page backward through the information. Press **Cancel** to exit from this service aid and return to the Service Aids menu.

Note: The appropriate key sequence for each function will appear at the bottom of the panel.

Display Previous Diagnostic Results

Note: This service aid is not available when you load stand-alone diagnostics from diskette.

This selection displays the results of previous diagnostic runs. Each time the diagnostics produce an SRN to report a problem, information about that problem is logged. The service representative can look at this log to see what SRNs are recorded. This log also records the results of diagnostic tests that run in loop mode. When you load this service aid, information on the last problem logged is displayed. By pressing **Enter** repeatedly, you can view up to 25 previous panels.

Display or Change Configuration or Vital Product Data (VPD)

This selection displays the Machine Configuration or VPD, and allows changes to the VPD.

Display Machine Configuration: This service aid displays a list of the resources installed on the system.

Display VPD: This service aid displays the VPD for all of the resources installed on the system.

Display or Alter VPD

Warning: If this service aid was loaded from a diskette, any changes or additions to the VPD will be lost when the system is shut down.

This service aid allows you to display and alter the VPD for any component. When you select this service aid, a menu allows you to select the desired component. When making a change to the VPD, the change must be saved by pressing the **Commit** key as defined at the bottom of the Display or Alter VPD panel.

Display or Change Diagnostic Test List

This selection displays or changes the resources tested by diagnostics during a POST. The diagnostic controller uses the diagnostic test list to determine which resources to test. This service aid provides a way to delete a resource from the diagnostics test list, a way to add a deleted resource back into the diagnostics test list, and a way to display the diagnostic test list.

Diskette Media

This selection provides a tool for verifying a diskette. When this service aid is selected, a menu asks you to select the type of diskette being verified. The program then reads all of the ID and data fields one time and displays the total number of bad sectors found.

Reset Bootlist

This selection provides a tool for clearing an incorrect bootlist. Incorrect bootlist is indicated by the IBM 6611 being unable to boot and, instead, halting with two numbers between **221** and **296** alternating on the 3-digit display or halting with a **260**, **261**, or **262** on the display.

Understanding the Cable Label

Each adapter cable has an IBM 6611 Cable Label at each end. The cable label provides information about the adapter and the network device to which the adapter cable is connected. See the following figure for a sample cable label.

Adapter Cable Connections	Network Management Information		
IBM 6611 Unit Number or Name: <i>1002</i> IBM 6611 Slot Number: <i>1</i>	Network Number or Name: CHQNET		
Adapter Type:Token-RingCAdapter Cable Type:Token-RingC	IBM 6611 ID at Network Manager: <i>6611ACHQ</i> IBM 6611 Address at Network: <i>10005A100001</i> <i>9.0.0.0</i>		
This Cable Connects To Device Type: <i>8228</i> Device Location: <i>CHQ/205, Rack 3, Raleigh</i> Device Number or Name: <i>MAU4</i>	Network Manager Location: <i>CHQ15, Main Site</i> Network Manager Operator's Phone Number: <i>(555) 555–5555</i>		

ACL2A046

Explanation of Label Fields

The IBM 6611 Cable Label has two major sections. The left half of the label, **Adapter Cable Connections**, provides information about the IBM 6611 adapter and the network device to which the adapter cable will be connected. This half of the label also assists with installation and problem determination.

The right half of the cable label, **Network Management Information**, provides information about the network to which the adapter will be attached and about the network management support, if any, for that network. This half of the label assists with problem determination.

IBM 6611 Unit Number or Name: This field contains the unique unit number or name that your enterprise assigned to the IBM 6611 to distinguish it from other IBM 6611s. The *IBM 6611 Network Processor Introduction and Planning Guide* instructs the customer to place a unit identification label on each IBM 6611 and record that information on the IBM 6611 Cabling Chart.

IBM 6611 Slot Number: This field contains the unique number of the IBM 6611 adapter slot in which the network adapter is installed.

Adapter Type: This field contains the type of network adapter that is installed in your machine. The *IBM 6611 Network Processor Introduction and Planning Guide* instructs the customer to use these types to represent the following adapters:

Adapter Type	Adapter Name
Token Ring	6611 Token-Ring Network 16/4 Adapter
	6611 Model 120 Token-Ring Network 16/4 Adapter
Ethernet	6611 Ethernet Adapter
	6611 Model 120 Ethernet Adapter
EIA 422/449 Serial	6611 2-Port EIA 422/449 Serial Adapter
	6611 Model 120 2-Port EIA 422/449 Serial Adapter
V.35/V.36 Serial	6611 2-Port V.35/V.36 Compatible Serial Adapter
	6611 Model 120 2-Port V.35/V.36 Compatible Serial Adapter
SDLC	6611 4-Port SDLC Adapter
X.25	6611 X.25 Adapter

Adapter Cable Type: This field contains the type of cable that must be connected to the network adapter.

Cable Type	Cable Name
Token Ring	6611 Token-Ring Network Adapter Cable
Ethernet	Ethernet Cable
EIA 422/449 Serial	6611 EIA 422/449 Serial Adapter Cable
V.35 Serial	6611 V.35 Compatible Serial Adapter Cable
V.35F Serial	Attachment Cable for V.35 DCE in France
V.36 Serial	6611 V.36 Compatible Serial Adapter Cable
SDLC IC	6611 SDLC Adapter Interface Cable
SDLC 232/V.24	6611 SDLC Adapter EIA 232/CCITT V.24 Cable
SDLC V.35	6611 SDLC Adapter CCITT V.35 Cable
SDLC X.21	6611 SDLC Adapter CCITT X.21 Cable Serial Adapter CCITT X.21 Cable
X.25 232/V.24	6611 X.25 Adapter EIA 232/CCITT V.24 Cable
X.25 X.21	6611 X.25 Adapter CCITT X.21 Cable
X.25 V.35	6611 X.25 Adapter CCITT V.35 Cable

This Cable Connects to Device Type: This field contains the type of device (such as a modem or an IBM 8228 Multistation Access Unit) to which the adapter cable must be connected.

This Cable Connects to Device Location: This field contains the physical location (such as the building, floor, wiring closet number, or rack number) of the device to which the adapter cable must be connected.

This Cable Connects to Device Number or

Name: This field contains the unique number or name that your enterprise assigned to the device to which the adapter cable must be connected. The unique number or name is used to identify the device.

Network Number or Name: This field contains the unique number or name assigned to the network segment to which the adapter will be attached. For example, in IBM Token-Ring Networks, each network segment (ring) has a unique number.

IBM 6611 ID at Network Manager: This field contains the name or address by which the IBM 6611 is known at the network manager. The network manager is a network management program that is running on a workstation attached to the network, and provides monitoring and control functions for the IBM 6611, or for individual adapters in the IBM 6611.

IBM 6611 Address at Network: This field contains the name or address that identifies this adapter to the network manager.

Network Manager Location: This field contains the physical location of the network manager. The customer may record the city, building number, or other details as desired.

Network Manager Operator's Phone Number:

This field contains the phone number of the operator at the network manager location.

Appendix A. 3-Digit Display Codes

Reading the 3-Digit Display



Figure A-1. An Error Code Displayed on the Model 120



Figure A-2. An Error Code Displayed on the Model 140



Figure A-3. An Error Code Displayed on the Model 170

The 3-digit display on the operator panel is used to:

- Track the progress of the IBM 6611 power-on self-test (POST) and other tests that run when the IBM 6611 is started
- Display the codes when the IBM 6611 software comes to an abnormal end
- Display the IBM 6611 messages
- Display diagnostic program codes.

During the POST, the numbers that are displayed indicate the progress of the testing. If an error is detected that requires attention, the IBM 6611 stops and a number is displayed on the 3-digit display to identify the error. When the POST runs without error, the 3-digit display is blank.

If you have a continuous flashing **888** on the 3-digit display, the IBM 6611 Network Processor Diagnostic Program has detected an error. See "Reading Flashing 888 Numbers" on page A-3 for instructions about reading the error codes.

Reading Flashing 888 Numbers

A flashing **888** on the 3-digit display of the IBM 6611 indicates that the IBM 6611 Network Processor Diagnostic Program has detected an error. Follow this procedure whenever a flashing **888** appears on the display.

- 1. Ensure that the key mode switch is set to *Normal* or *Service*.
- 2. Press **Reset** and hold for about 1 second to allow the program to sense the change.
- Record the number on the 3-digit display. This number indicates the message type.
- 4. Proceed according to the message type:

Туре 102:	See "Reading Type 102 Messages."
Туре 103:	See "Reading Type 103 Messages" on page A-4.
Other types:	See "Reading Other Message Types" on page A-6.

Reading Type 102 Messages

Follow this procedure to determine the crash code and the dump status code associated with type **102** messages.

- Press Reset once. Record the crash code. See "Crash Codes for Type 102 Messages" on page A-7 to interpret the code.
- Press Reset again. Record the dump status code. See "Dump Status Codes for Type 102 Messages" on page A-8 to interpret the code.
- Press Reset again and note whether a flashing 888, or a steady 103, or a steady ccc is on the 3-digit display.
 - If a type 103 message is displayed, go to "Reading Type 103 Messages" on page A-4.

- If a type ccc message is displayed, another message will follow. Go to Step 2 under "Reading Flashing 888 Numbers."
- If a flashing 888 is displayed, all of the information about the type 102 message has been displayed. If you want to repeat the message sequence, return to Step 1 on page A-3 of this procedure.

You must restart the IBM 6611 to recover from this step.

Reading Type 103 Messages

Follow this procedure to determine the service request number (SRN) and the field replaceable unit (FRU) location codes associated with type 103 messages.

- 1. Press **Reset** once and record the number on the 3-digit display. This number is the first 3 digits of the 6-digit SRN.
- 2. Press **Reset** again and record the second 3 digits of the 6-digit SRN.
- 3. You must read the FRU location code associated with this type 103 message.
 - There may be multiple FRU location codes associated with each type 103 message. However, no more than 4 FRU location codes will be displayed.
 - Each FRU location code is preceded by an identifier (c0x) that indicates whether it is the first, second, third, or fourth FRU location code.
 - Each FRU location code can be up to 24 digits long and is displayed as containing up to eight 3-digit numbers.

This is an example of a sequence of 4 FRU location codes:

c01 1xx 2xx 3xx 4xx 5xx 6xx 7xx 8xx=First FRU c02 1xx 2xx 3xx 4xx 5xx 6xx 7xx 8xx=Second FRU c03 1xx 2xx 3xx 4xx 5xx 6xx 7xx 8xx=Third FRU c04 1xx 2xx 3xx 4xx 5xx 6xx 7xx 8xx=Fourth FRU

To read the FRU location codes:

- a. Press **Reset** once and **c0x** will be displayed.
- Press **Reset** again and record the first 3 digits of the 24-digit FRU location code. Repeat this step until you have recorded all 24 digits.
- c. Press Reset again:
 - If c0x is displayed, return to Step 3b to record the next FRU location code.
 - If ccc is displayed, return to Step 1 on page A-4 to read the next type 103 message.
 - If a flashing 888 is displayed, there are no more FRU location codes or type 103 messages. Continue with Step 4 on page A-6.
- d. You must determine the 8-digit location code (AB-CD-EF-GH) associated with the FRU location code. Each digit of the 8-digit location code is presented as a 3-digit number on the 3-digit display. The following example shows the relationship between the 8-digit location code and the FRU location code:

	Α	В	С	D	Е	F	G	H =8-digit
c01	1xx	2xx	3xx	4xx	5xx	6xx	7xx	8xx=First FRU
c02	1xx	2xx	3xx	4xx	5xx	бхх	7xx	8xx=Second FRU
c03	1xx	2xx	3xx	4xx	5xx	6xx	7xx	8xx=Third FRU
c04	1xx	2xx	3xx	4xx	5xx	6xx	7xx	8xx=Fourth FRU

١

If a **ccx** (x can be any digit from 2 through 9) is displayed as part of the location code, only the part of the code that is different from the location code of the previous FRU is shown. To form the complete location code of the next FRU, substitute the information following the ccx into the location code of the previous FRU. For example, if the previous FRU location is: 100 200 300 401 500 601 700 c01 800 and the next FRU location is listed as:

cc2 602

the complete location code of the next FRU is: cc2 100 200 300 401 500 602 700 800. To identify each digit of the location code (AB-CD-EF-GH), translate the rightmost 2 digits using the following table.

xx	Value	xx Value	xx Value	xx Value
00	= 0	11 = A	21 = K	31 = U
01	= 1	12 = B	22 = L	32 = V
02	= 2	13 = C	23 = M	33 = W
03	= 3	14 = D	24 = N	34 = X
04	= 4	15 = E	25 = O	35 = Y
05	= 5	16 = F	26 = P	36 = Z
06	= 6	17 = G	27 = Q	
07	= 7	18 = H	28 = R	
08	= 8	19 = 1	29 = S	
09	= 9	20 = J	30 = T	

4. To recover from the IBM 6611 stop indicated by the flashing 888, you must have the customer restart the IBM 6611.

Reading Other Message Types

The only valid message types are Type 102, 103, and 104. Type 104 messages may be ignored. If another 3-digit number is displayed:

- 1. Press **Reset** until a flashing **888** appears on the 3-digit display.
- 2. When a flashing **888** appears, go to "Reading Flashing 888 Numbers" on page A-3.
- If a flashing 888 does not appear, you should consider the numbers as steady numbers. Record 101-xxx where xxx is the steady number displayed on the 3-digit display.

Crash Codes for Type 102 Messages

This section contains a list of the crash codes that are part of a type 102 message. See "Reading Type 102 Messages" on page A-3 for instructions about how to read a crash code from the 3-digit display.

The crash codes are:

- 000 Unexpected system interrupt.
- 200 System check because of a memory bus error.
- 201 System check because of a memory timeout.
- 202 System check because of a memory card failure.
- **203** System check because of an address that is out of range.
- 204 System check because of an attempt to write to ROS.
- 205 System check because of an unrecoverable address parity.
- 206 System check because of an unrecoverable ECC error.
- 207 System check because of an unidentified error.
- 300 Data storage interrupt from the processor.
- **32x** Data storage interrupt because of an I/O exception from IOCC.
- **38x** Data storage interrupt because of an I/O exception from SLA.

400	Instruction storage interrupt.
500	External interrupt because of a scrub memory bus error.
501	External interrupt because of a DMA memory bus error.
52x	External interrupt because of an IOCC channel check.
53x	External interrupt because of an IOCC bus timeout.
54x	External interrupt because of an IOCC connector check.
558	Not enough memory space to continue.
700	Program interrupt.
800	Floating point is not available.

Dump Status Codes for Type 102 Messages

This section contains a list of the dump status codes that are part of a type 102 message. Dump status codes are displayed on the 3-digit display as explained in "Reading Type 102 Messages" on page A-3.

The leftmost position of each of the dump status codes is blank and a 0 is used in the list to represent the blank. A lowercase c is displayed in the lower half of the character position on the display.

The dump status codes are:

- 0c0 The dump completed successfully.
- 0c2 A user-requested dump has started.
- 0c3 The dump is inhibited.
- **0c4** The dump has not been completed. A partial dump may be present.
- **0c5** The dump program could not access the dump device.

- 0c7 Reserved.
- 0c8 The dump function is disabled.
- 0c9 A dump is in progress.
- c20 System debug program is activated.

See "Error and Status Codes" for additional information.

Error and Status Codes

This section lists the error and status codes that can appear on the 3-digit display of the IBM 6611. The list is arranged in sequential order, not in hexadecimal order. For a detailed list of the Error and Status Codes, refer to the *Multiprotocol Network Program User's Guide*.

Code 0c0

Explanation: A user-requested dump has been completed successfully.

User Response: Progression indicator; no action.

Code 0c2

Explanation: A user-requested dump has started.

User Response: Progression indicator; no action.

Code 0c3

Explanation: The dump is inhibited.

User Response: The dump procedure cannot be performed at this time. Regenerate the failing operational scenario and perform the dump procedure correctly once the operational problem has been re-created.

Code 0c4

Explanation: The dump has not been completed. A partial dump may be present.

User Response: Check the dump device for partial dump output. Retrieve the partial dump output. Regenerate the failing operational scenario and perform the dump procedure correctly once the operational problem has been re-created.

Code 0c5

Explanation: The dump has failed to start. The dump program did not write to the dump device.

User Response: Check the diskette drive for correct diskette format. Regenerate the failing operational scenario and perform the dump procedure correctly once the operational problem has been re-created.

Code 0c8

Explanation: The dump is disabled. The system configuration does not include a dump device for the requested dump.

User Response: Configure and specify the dump device. Regenerate the failing operational scenario and perform the dump procedure correctly once the operational problem has been re-created.

Code 0c9

Explanation: A system-initiated dump is in progress. An unexpected system halt occurred and a system dump has started automatically.

User Response: Progression indicator; no action.

Code 10c-15c, 20c-25c, 30c-35c, 40c-45c, 50c-55c, 60c-65c, 70c-75c

Explanation: The system monitor process cannot access either the adapters installed in slots 1-7 (respective to the leftmost digit) or the protocol function code (respective to the middle digit) due to a problem with:

- · System code that communicates with the adapters
- Adapter microcode
- Adapter hardware.

User Response: For suspected software problem (CE):

Check error log to determine software problem.

For suspected hardware problem (CE):

- Execute the adapter diagnostics
- · Check the adapters for correct installation
- · Check the adapters for defects.

Code 1c1, 1c3, 1c5, 1c7, 1c9

Explanation: Starting protocol and configuration.

User Response: Progression indicator; No action.

Code 1c2, 1c4, 1c6, 1c8, 4c1, 4c6

Explanation: System monitor cannot start the IP configuration error log or the routing process due to the following representative errors:

- Process crash
- Process halted waiting for event trigger
- Process start threshold exceeded
- Process using too much memory
- Process using too much CPU time.

User Response: Progression Indicator; protocol disabled upon IPL completion.

Code 2c0

Explanation: Software is locked.

User Response: Progression indicator. IPL is halted. Software must be unlocked before successful IPL completion. Verify configuration and check with your system administrator.

Code 2c1-2c9, 3c0-3c9, 4c0, 4c2, 4c3, 4c4, 4c5, 4c7-4cc

Explanation: Software lock check and IPL process initiation. The following processes are initiated:

- IP protocol
- XNS protocol
- DECnet protocol
- SNMP protocol
- DLS protocol
- System monitor
- Process monitor
- Disk monitor
- Protocol monitor
- Interface monitor
- IPX protocol
- X.25
- Smart Card Routing Adapter Management
- Point-to-Point Protocol
- Token-Ring Bridging
- Transparent Bridging
- Appletalk protocol
- APPN program
- Asynchronous Routing Communications.

User Response: Progression indicator; no action.

Code 5c5

Explanation: The system monitor process cannot start the IBM Token-Ring Network Tranparent Bridge Program process due to the following representative errors:

- Process crash
- Process hung waiting for event trigger
- Process start threshold exceeded
- Process using too much memory
- Process using too much CPU time.

User Response: Progression Indicator, The transparent bridge spanning tree function is disabled upon IPL completion.

Code 5c6

Explanation: The system monitor process cannot start the Appletalk protocol process due to the following representative errors:

- Process crash
- · Process hung waiting for event trigger
- · Process start threshold exceeded
- Process using too much memory
- Process using too much CPU time.

User Response: Progression Indicator. The Appletalk protocol is disabled upon IPL completion.

Code 5c7

Explanation: The system monitor process cannot start the asynchronous routing communication process due to the following representative errors:

- Process crash
- Process hung waiting for event trigger
- Process start threshold exceeded
- · Process using too much memory
- · Process using too much CPU time.

User Response: Progression Indicator. The asynchronous routing communication process is disabled upon IPL completion.

Code 80c-99c, c40-c43

Explanation: The system monitor process cannot start functional monitoring and SNMP subagent processes due to the following representative errors:

- Process crash
- Process halted waiting for event trigger
- Process start threshold exceeded
- Process using too much memory
- Process using too much CPU time.

User Response: Progression indicator. The processes are disabled upon IPL completion.

Code c20

Explanation: The system debug program is activated due to an unexpected system halt that has occurred.

User Response: Progression indicator; flashing **888** is displayed and a system dump is performed after **c20** is displayed.

Code c50-c61

Explanation: Shutdown progression indicators.

User Response: Progression indicator; no action.

Code c67-c71

Explanation: Configuration diskette errors during IPL or code updates.

User Response: Progression indicator; the configuration process temporarily waits for problem resolution. Insert a valid configuration diskette into the diskette drive. For initial configuration, IPL operation continues if the diskette is inserted within 1 minute; after 1 minute, reinitiate the IPL operation with the configuration diskette in the diskette drive. For subsequent configurations, normal operation continues with the previous configuration active until the new configuration diskette is installed and the update sequence invoked from the System Manager.

Code c72

Explanation: Configuration errors detected after the configuration file is received. Representative errors include:

- Incorrect configuration data in file
- · Configuration data and hardware mismatch
- Function validity
- Function interdependency error.

User Response: Check the error log for error details. Resolve the configuration error. For initial configuration, reinitiate the IPL procedure with the corrected configuration. For subsequent configurations, normal operation continues with the previous configuration active until the new configuration is available (configuration errors are resolved).

Code c73

Explanation: Wrong Configuration Program version number. The version of the configuration program must match the version of the Multiprotocol Network Program.

User Response: Acquire the correct Configuration Program version and ensure that it matches the Multiprotocol Network Program version. Reconfigure if necessary. Apply matching Configuration Program version to IBM 6611 with matching Multiprotocol Network Program version. Normal operation continues with the previous configuration active until the configuration and code versions match.

Code c74

Explanation: Wrong IBM 6611 model number. The model number in the configuration file does not match the model number of the IBM 6611.

User Response: Reconfigure the model number in the configuration. For initial configuration, reinitiate the IPL operation after the model number configured matches the machine model to which it is being applied. For subsequent configurations, normal operation continues with the previous configuration active until the configuration model and machine models match (new configuration file is available).

Code c80

Explanation: 6611 is activating all configured network adapters.

User Response: Progression indicator; no action.

Code c81

Explanation: The configuration data that requires the machine to be reinitialized (IPL) has changed. The IPL is in progress.

User Response: Progression indicator; no action.

Code c89

Explanation: The configuration process is complete. The adapters are initialized.

User Response: Progression indicator; no action.

Code c99

Explanation: Diagnostics are complete.

User Response: Progression indicator; no action.

This code is used only when there is no terminal attached to serial port 1.

Appendix B. Ordering Keys

For protection against unauthorized key duplication, the key mode switch is equipped with a Medeco high-security lock. Keys for this lock are a factory-restricted series, and duplicate keys *are not* available through normal commercial channels. The metal code tag supplied with your original keys authorizes you to purchase additional keys directly from the Medeco factory. The additional key supplied and the metal code tag should be stored in a secured area.

To obtain information or replacement keys, use the list below to contact the nearest Medeco distributor. Complete a copy of the order form and mail it to the distributor. As a security precaution, Medeco will not accept orders that do not include both the code tag and the official order form.

USA

Medeco Department KLC P.O. Box 3075 Salem, VA 24153

Europe

Claus Clausen 89a Authur Road Wimbledon Park, London SW 19 7DP England Phone: 011-44-81-946-2823 Fax: 011-44-81-946-2286

Far East

Sung Kim Geoho Corporation P.O. Box 519 Kwang Chan Bldg. 4th Floor 983-42 Bangbia-dong Seocha-Ku, Seoul Korea Phone: 011-82-02-521-2100 Fax: 011-82-02-521-2106

Australia

Neville Burr ATM Lock A Safe Co. Pty. Ltd. 2/48 Ourimbah Road, P.O. Box 300 Tweed Head, N.S.W. 2485 Australia Phone: 011-61-075-36-1611 Fax: 011-61-075-36-1611

Latin America

Ricardo DeCastro Calle 22 No 3-30, Ofc. 201 P.O. Box A.A. No. 39955 Bogota, Columbia Sur America Phone: 011-57-1-268-5827 or 6180 Fax: 011-57-1-268-2628

Middle East

Moshe Rotner R.M. Rotan Marketing 34 Nordau Street, Herzlia B P.O. Box 5138, Herzlia Israel Phone: 011-972-52-504622 Fax: 011-972-52-582357

Key Reorder Form

A copy of this form, when accompanied by the metal code tag supplied with the original keys, represents an IBM authorized order for additional factory keys.

Please indicate the quantity required and enclose a check or money order for the appropriate amount.

Number of keys required _____

Please Type or Print Your Return Address

Name	
Address	
City	
State	Zip
Country	

Select an address from the list provided on the previous page, and mail a copy of this form to that location.

Your key code tag will be returned with your new keys.

Note: No orders will be processed without both the key tag and this form.

ACL0A010

B-4 6611 Operations Pocket Guide

List of Abbreviations

ACSU	Advanced Channel Service Unit
AIX	Advanced Interactive Executive
ASCII	American National Standard Code for Information Interchange
АТ	Attention command set
bps	bits per second
CCITT	Consultative Committee on International Telegraph and Telephone
CE	Customer Engineer
CPU	central processing unit
CR	carriage return
CRT	cathode ray tube
CSR	configuration report server
DLS	data link switching
DMA	direct memory access
ECC	error correction circuitry
EIA	Electronics Industries Association
ETX	end of text
FDX	full duplex
FRU	field replaceable unit
ID	identification
I/O	input/output
1000	input/output channel controller
IP	Internet Protocol
IPL	initial program load
IPRTS	permanent request to send
IPX	Internet Packet Exchange
LED	light-emitting diode
LF	line feed

MB	megabyte (1048576 bytes)
ms	millisecond
PC	personal computer
POST	power-on self-test
PRTS	permanent request to send
RAM	random access memory
RISC	reduced instruction-set computer
ROS	read-only storage
SCSI	small computer systems interface
SDLC	synchronous data link control
SLA	serial link adapter
SNMP	Simple Network Management Protocol
SRN	service request number
ТСР	Transmission Control Protocol
VPD	vital product data

Reader's Comment

IBM 6611 Network Processor Operations Pocket Guide

Publication No. GX27-3909-02

Use this form to tell us what you think about this manual. If you have found errors in it, or if you want to express your opinion about it (such as organization, subject matter, appearance) or make suggestions for improvement, this is the form to use.

To request additional publications, or to ask questions or make comments about the functions of IBM products or systems, you should talk to your IBM representative or to your IBM authorized remarketer. This form is provided for comments about the information and the way it is presented.

When you send information to IBM, you grant IBM a nonexclusive right to use or distribute the information in any way it believes appropriate without incurring any obligation to you.

Be sure to print your name and address below if you would like a reply.

Comments:

Name:	
Company Name:	
Address:	
Phone No [.]	

NO POSTAGE NECESSARY IF MAILED IN THE UNITED STATES

BUSINESS REPLY MAIL

FIRST CLASS MAIL PERMIT NO. 40 ARMONK, NEW YORK

POSTAGE WILL BE PAID BY ADDRESSEE

International Business Machines Corporation Information Development and Design Department E15 P.O. Box 12195 Research Triangle Park, North Carolina 27709-9990

Indiation BrootBrootstaal daala baala daalka adaalad



Part Number: 34G2659

Printed in U.S.A



